Author's response to reviews

Title: Interventions to reduce neonatal mortality from neonatal tetanus in low and middle income countries - a systematic review

Authors:

Adeel Ahmed Khan Dr (adeel.khan@aku.edu)
Aysha Zahidie Dr (aysha.zahidie@aku.edu)
Fauzia Rabbani Dr (fauzia.rabani@aku.edu)

Version: 4 Date: 15 June 2012

Author's response to reviews:

Dear Editor

Great Day

I have selected this journal for the publication of our article because this journal is publishing research widely in the field of Public health and we have found this journal correspondence quite professional and feasible from authors point of view.

Regards

Dr Aysha Zahidie